Author's response to reviews

Title: TECNOB: study design of a randomized controlled trial of a multidisciplinary telecare intervention for obese patients with type-2 diabetes.

Authors:

Gianluca Castelnuovo (gianluca.castelnuovo@auxologico.it)
Gian Mauro Manzoni (gm.manzoni@auxologico.it)
Paola Cuzziol (p.cuzziol@auxologico.it)
Gian Luca Cesa (gluca.cesa@gmail.com)
Cristina Tuzzi (c.tuzzi@auxologico.it)
Valentina Villa (v.villa@auxologico.it)
Antonio Liuzzi (liuzzi@auxologico.it)
Maria Letizia Petroni (ml.petroni@auxologico.it)
Enrico Molinari (molinari@auxologico.it)

Version: 2 Date: 23 February 2010

Author's response to reviews:

Gianluca Castelnuovo Ph.D.,Psy.D.
Clinical Psychology Lab
San Giuseppe Hospital
IRCCS Istituto Auxologico Italiano, Verbania, Italy
gianluca.castelnuovo@auxologico.it
Tel. - Fax. +39-0323514338

Faculty of Psychology
Catholic University of Milan
gianluca.castelnuovo@unicatt.it
Tel. +39-3493257042

February 23, 2010

Reply to Referee

Thank you for your comments.

1. In the sample size section we stated and justified the sidedness of the test of hypothesis for the primary outcome as suggested.

2. In the analysis section we stated the sidedness and level of the test of hypotheses for all the secondary outcomes

3. We also addressed the issue of multiple comparisons.
All modifications are red-typed in the text.
Thank you for your time and consideration.
I look forward to hearing from you.
Sincerely,
Gianluca Castelnuovo